Author's response to reviews

Title: Brain metastases from breast cancer: prognostic significance of HER-2 overexpression, effect of trastuzumab and cause of death

Authors:

Romuald Le Scodan (lescodan@crh1.org)
Ludivine Jouanneau (ludivine.jouanneau@curie.net)
Christophe Massard (christophe.massard@igr.fr)
Maya Gutierrez (maya.gutierrez@curie.net)
Youlia Kirova (youlia.kirova@curie.net)
Pascal Cherel (pascal.cherel@curie.net)
Julie Gachet (jgachet@hotmail.fr)
Alain Labib (alain.labib@curie.net)
Emmanuelle Mouret-Fourme (emmanuelle.fourme@curie.net)

Version: 6 Date: 26 August 2011

Author's response to reviews:

corrections have been made